Author's response to reviews

Title: Recurrence of hepatitis C virus during agranulocytosis and spontaneous clearance after recovery from cytopenia

Authors:

Norbert H Gruener (Norbert.Gruener@med.uni-muenchen.de)
Bijan Raziorrouh (Bijan.Raziorrouh@med.uni-muenchen.de)
Maria-Christina Jung (Maria-Christina.Jung@med.uni-muenchen.de)

Version: 2 Date: 2 September 2007

Author's response to reviews:

Subheading 'Background' has been changed to 'Introduction'.
Title ends with: a case report
Keywords section has been removed